Author's response to reviews

Title: Retroperitoneoscopic single-site renal biopsy surgery: right indications for the right technique

Authors:
Salvatore Micali (salvatore.micali@unimore.it)
Alessio Zordani (alessio_zordani@hotmail.it)
Riccardo Galli (gallidoc@gmail.com)
Eugenio Martorana (eugeniomartorana@libero.it)
Micaela Piccoli (m.piccoli@ausl.mo.it)
Gianni Cappelli (gianni.cappelli@unimore.it)
Giampaolo Bianchi (bianchi.giampaolo@unimo.it)

Version: 2
Date: 19 May 2014

Author's response to reviews:

Dear Editors,

I modified the manuscript following Your instructions,

1) and 2) 3)

I change the type of manuscript with technical advance section

3) Requesting for Competing Interest Section:

Please provide us with a Competing Interest Section.

4) Authors' contributions:

I modified the manuscript with the Authors' contributions section

5) Line numbering:

Now the manuscript include line numbers

6) Email Address:

I modified the manuscript with the email addresses of all authors in the title page.
I have the same pictures in RGB modality. If you prefer I can change the
pictures.